Author’s response to reviews

Title: Inequities in coverage of smokefree space policies within the United States

Authors:

Christopher Lowrie (lowriechris779@gmail.com)
George Thomson (george.thomson@otago.ac.nz)
Amber Pearson (apearson@msu.edu)

Version: 2 Date: 21 Mar 2017

Author’s response to reviews:

The Editor, BMC Public Health

Dear Editor,

Thank you for your clarifications on formatting. We have made the suggested revisions, and now resubmit for your consideration.

Editor Comments:

Comment 1:

Please format the abstract in your ms according to the guidelines.

The Abstract has been reformatted to include subsections, with minor rewording to accommodate this formatting.

Comments 2 & 3:

Please provide a 'Declarations' section heading after your conclusions. Please provide a full Declarations section as detailed in the guidelines.

A Declarations section is provided, with appropriate subsections included.
Thank you,

Christopher J. Lowrie, on behalf of all authors